Author's response to reviews

Title: Effects of EpCAM Overexpression on Human Breast Cancer Cell Lines

Authors:

Johanna M Gostner (jgostner@gmail.com)
Dominic Fong (dominic.fong@i-med.ac.at)
Oliver A Wrulich (oliver.wrulich@i-med.ac.at)
Florian Lehne (florian.lehne@gmail.com)
Marion Zitt (marion.zitt@i-med.ac.at)
Martin Hermann (martin.hermann@i-med.ac.at)
Sylvia Krobutsch (krobutsch@molgen.mpg.de)
Agnieszka Martowicz (agnieszka.martowicz@i-med.ac.at)
Guenther Gastl (Guenther.Gastl@i-med.ac.at)
Gilbert Spizzo (Gilbert.Spizzo@i-med.ac.at)

Version: 5 Date: 24 January 2011

Author's response to reviews: see over
Dear Editor,

thank you for the opportunity to publish our manuscript in BMC Journal.

Here are the formatting changes that we have performed point-by point.

Major revisions

„Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution“

➔ The affiliations have now been corrected.

„Abbreviations: Please include these as a paragraph of text, not a tables“

➔ A new paragraph has been inserted.

„Figures: Please provide a single figure title for each figure. These should be concise and be the first sentence of the figure legend。“

➔ Figure Titles have now been inserted.

„Tables: Please ensure that both tables are cited in the manuscript in numerical order. Table 1 is not currently cited.“
⇒ Table 1 has now been cited in the text

Minor revisions

„Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.“

⇒ Highlighting and tracking have been removed

„Tables: please remove the visible vertical lines from your tables“

⇒ vertical lines have been removed

„Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.“

⇒ We have cropped and corrected the Figures. Please check if they are OK.

„Figures: Please ensure all mention of figures is consistent (it should be Figure or Fig but not a mixture of both)“

⇒ The word „Figure“ has been used throughout the manuscript

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.

⇒ We have checked and corrected typographical errors.

With best regards
Gilbert Spizzo and Johann Gostner